Gulfs of Execution and Evaluation

CS 4163

Summer 2002

The gulfs are described in your textbook [4, p.273].

Figure 1: Bridging The Gulfs (From a chapter by Norman [2, Figure 3.2, p. 40])

References

[1] Edwin L. Hutchins, James D. Hollan, and Donald A. Norman. Direct manipulation interfaces. In Norman and Draper [3], chapter 5, pages 87 – 124.

[2] Donald A. Norman. Cognitive engineering. In Norman and Draper [3], chapter 3, pages 31 – 61.

[3] Donald A. Norman and Stephen W. Draper, editors. User Centered System Design: New Perspectives on Human-Computer Interaction. Lawrence Erlbaum Associates, Publishers, Hillsdale, New Jersey and London, 1986. LCC=QA76.9.I58U73, ISBN=0-89859-781-1 (hc).

[4] Jenny Preece, Yvonne Rogers, Helen Sharp, David Benyon, Simon Holland, and Tom Carey. Cognitive issues in direct manipulation. In Human-Computer Interaction, section 13.7, pages 273 – 282. Prentice-Hall Europe, 1994. ISBN 0-201-62769-8.

Figure 2: Norman’s Activity Model (From a chapter by Norman [2, Figure 3.3, p. 42])

Figure 3: Incorporating Norman’s Activity Model Into The Gulfs (From a chapter by Hutchins et al. [1, Figure 5.7,p. 111])